% EJC papers *must* begin with the following two lines. 
\documentclass[12pt]{article}
\usepackage{e-jc}
\specs{P2.12}{24(2)}{2017}


% Please remove all other commands that change parameters such as
% margins or pagesizes.

% only use standard LaTeX packages
% only include packages that you actually need

% we recommend these ams packages
\usepackage{amsthm,amsmath,amssymb}

% we recommend the graphicx package for importing figures
\usepackage{graphicx}

% use this command to create hyperlinks (optional and recommended)
\usepackage[colorlinks=true,citecolor=black,linkcolor=black,urlcolor=blue]{hyperref}

% use these commands for typesetting doi and arXiv references in the bibliography
\newcommand{\doi}[1]{\href{http://dx.doi.org/#1}{\texttt{doi:#1}}}
\newcommand{\arxiv}[1]{\href{http://arxiv.org/abs/#1}{\texttt{arXiv:#1}}}

% all overfull boxes must be fixed; 
% i.e. there must be no text protruding into the margins


% declare theorem-like environments
\theoremstyle{plain}
\newtheorem{theorem}{Theorem}
\newtheorem{lemma}[theorem]{Lemma}
\newtheorem{corollary}[theorem]{Corollary}
\newtheorem{proposition}[theorem]{Proposition}
\newtheorem{fact}[theorem]{Fact}
\newtheorem{observation}[theorem]{Observation}
\newtheorem{claim}[theorem]{Claim}

\theoremstyle{definition}
\newtheorem{definition}[theorem]{Definition}
\newtheorem{example}[theorem]{Example}
\newtheorem{conjecture}[theorem]{Conjecture}
\newtheorem{open}[theorem]{Open Problem}
\newtheorem{problem}[theorem]{Problem}
\newtheorem{question}[theorem]{Question}

\theoremstyle{remark}
\newtheorem{remark}[theorem]{Remark}
\newtheorem{note}[theorem]{Note}


\def\bA{\mathbf A}
\def\bE{\mathbf E}
\def\bF{\mathbf F}
\def\bG{\mathbf G}
\def\bP{\mathbf P}
\def\bN{\mathbf N}
\def\bZ{\mathbf Z}
%\def\K{K}
\def\bL{\mathbf L}
\def\bQ{\mathbf Q}
\def\bU{\mathbf U}
\def\bV{\mathbf V}
\def\bW{\mathbf W}
\def\bX{\mathbf X}
\def\bY{\mathbf Y}
%
\def\cC{\mathcal C}
\def\cD{\mathcal D}
\def\cE{\mathcal E}
\def\cF{\mathcal F}
\def\cG{\mathcal G}
\def\cH{\mathcal H}
\def\cL{\mathcal L}
\def\cM{\mathcal M}
\def\cK{\mathcal K}
\def\cO{\mathcal O}
\def\cP{\mathcal P}
\def\cQ{\mathcal Q}
\def\cX{\mathcal X}
\def\cY{\mathcal Y}
\def\cU{\mathcal U}
\def\cV{\mathcal V}
\def\cT{\mathcal T}
\def\cR{\mathcal R}
\def\cS{\mathcal S}
\def\cW{\mathcal W}
\def\cFR{\mathcal{Fr}}
%
\def\PG{{\rm PG}}
\def\PSp{{\rm PSp}}
\def\PGL{{\rm PGL}}
\def\GF{{\rm GF}}
\def\AG{{\rm AG}}
\def\GL{{\rm GL}}
\def\PGO{{\rm PGO}}
\def\PGU{{\rm PGU}}


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

% if needed include a line break (\\) at an appropriate place in the title

\title{\bf Maximal partial spreads of polar spaces}

% input author, affilliation, address and support information as follows;
% the address should include the country, and does not have to include
% the street address 

\author{Antonio Cossidente\\ 
\small  Dipartimento di Matematica Informatica ed Economia\\[-0.8ex]
\small Universit\`a degli Studi della Basilicata \\[-0.8ex] 
\small Contrada Macchia Romana, I-85100 Potenza, Italy.\\
\small\tt antonio.cossidente@unibas.it\\
\and
 Francesco Pavese \\
\small Dipartimento di Meccanica Matematica e Management\\[-0.8ex]
\small Politecnico di Bari\\[-0.8ex]
\small Via Orabona, 4, I-70125 Bari, Italy.\\
\small\tt francesco.pavese@poliba.it
}

% \date{\dateline{submission date}{acceptance date}\\
% \small Mathematics Subject Classifications: comma separated list of
% MSC codes available from http://www.ams.org/mathscinet/freeTools.html}

\date{\dateline{Aug 20, 2015}{Apr 4, 2017}{Apr 13, 2017}\\
\small Mathematics Subject Classifications: 51A40, 51A50, 51E14, 51E23}

\begin{document}
\begin{thebibliography}{99}

\bibitem{ACE} A.~Aguglia, A.~Cossidente, and G.~L. Ebert. \newblock Complete spans on Hermitian varieties. \newblock {\em Des. Codes Cryptogr.}, 29:7--15, 2003.

\bibitem{MA} M.~Aschbacher. \newblock On the maximal subgroups of the finite classical groups. \newblock {\em Invent. Math.} 76(3):469--514, 1984.

\bibitem{BCL} L.~Bader, A.~Cossidente, and G.~Lunardon. \newblock Generalizing flocks of ${\cal Q}^+(3,q)$. \newblock {\em Adv. Geom.} 1:323--331, 2001.

\bibitem{BGS}  S.~Ball, P.~Govaerts, and L.~Storme. \newblock On ovoids of parabolic quadrics. \newblock {\em Des. Codes Cryptogr.} 38(1):131--145, 2006.

\bibitem{CS} A.~Cossidente, and L.~Storme. \newblock Caps on elliptic quadrics. \newblock {\em Finite Fields Appl.} 1(4):412--420, 1995.

\bibitem{BM} J.~De Beule, and K.~Metsch. \newblock The maximum size of a partial spread in $\cH(5,q^2)$ is $q^3+1$. \newblock {\em J. Combin. Theory Ser. A}, 114:761--768, 2007.

\bibitem{Gill} N.~Gill. \newblock Polar spaces and embeddings of classical groups. \newblock {\em New Zealand J. Math.} 36:175--184, 2007.

\bibitem{H}  J.~W.~P. Hirschfeld. \newblock {\em Finite projective spaces of three dimensions}. \newblock Oxford Mathematical Monographs, Oxford Science Publications, The Clarendon Press, Oxford University Press, New York, 1985.

\bibitem{HT} J.~W.~P. Hirschfeld, and J.~A. Thas. \newblock {\em General Galois Geometries}. \newblock Oxford Mathematical Monographs, Oxford Science Publications, The Clarendon Press, Oxford University Press, New York, 1991.

\bibitem{KL} P.~Kleidman, and M.~Liebeck. \newblock {\em The Subgroup Structure of the Finite Classical Groups}. \newblock Cambridge University Press, Cambridge (1990).

\bibitem{KMS} A.~Klein, K.~Metsch, and L.~Storme. \newblock Small maximal partial spreads in classical finite polar spaces. \newblock {\em Adv. Geom.} 10(3):379--402, 2010.

\bibitem{L} D.~Luyckx. \newblock On maximal partial spreads of $\cH(2n+1,q^2)$. \newblock {\em Discrete Math.} 308:375--379, 2008.

\bibitem{Grassl} P.~Mandayam, S.~Bandyopadhyay, M.~Grassl, and W.~K. Wootters. \newblock Unextendible mutually unbiased bases from Pauli classes. \newblock {\em Quantum Inf. Comput.} 14 (9-10):823--844, 2014.

\bibitem{S} B.~Segre. \newblock Forme e geometrie hermitiane con particolare riguardo al caso finito. \newblock {\em Ann. Mat. Pura Appl.} 70:1--201, 1965.

\bibitem{S1} B.~Segre. \newblock Teoria di Galois, fibrazioni proiettive e geometrie non desarguesiane. \newblock {\em Ann. Mat. Pura Appl.} 64:1--76, 1964.

\bibitem{T1} J.~A. Thas. \newblock Maximal exterior sets of hyperbolic quadrics: the complete classification. \newblock {\em J. Combin. Theory Ser. A} 56(2):303--308, 1991. 

\bibitem{T} J.~A. Thas. \newblock Old and new results on spreads and ovoids of finite classical polar spaces. \newblock Combinatorics '90 (Gaeta, 1990), {\em Ann. Discrete Math.} 52:529--544, 1992. 

\bibitem{KT} K.~Thas. \newblock Unextendible mutually unbiased bases (after Mandayam, Bandyopadhyay, Grassl and Wootters). \newblock {\em Entropy} 18(395):15pp., 2016.

\bibitem{V1} F.~Vanhove. \newblock The maximum size of a partial spread in $\cH(4n+1,q^2)$ is $q^{2n+1}+1$. \newblock {\em Electron. J. Combin.}, 16:1--6, 2009.

\bibitem{V2} F.~Vanhove. \newblock A geometric proof of the upper bound on the size of partial spreads in $\cH(4n+1,q^2)$. \newblock {\em Adv. Math. Commun.} 5:157--160, 2011.

\end{thebibliography}


\maketitle



\begin{abstract}
Some constructions of maximal partial spreads of finite classical polar spaces are provided. In particular we show that, for $n \ge 1$, $\cH(4n-1,q^2)$ has a maximal partial spread of size $q^{2n}+1$, $\cH(4n+1,q^2)$ has a maximal partial spread of size $q^{2n+1}+1$ and, for $n \ge 2$, $\cQ^+(4n-1,q)$, $\cQ(4n-2,q)$, $\cW(4n-1,q)$, $q$ even, $\cW(4n-3,q)$, $q$ even, have a maximal partial spread of size $q^n+1$.

  % keywords are optional
  \bigskip\noindent \textbf{Keywords:} finite classical polar space, maximal partial spread, Singer cycle, Segre variety
\end{abstract}





 \section{Introduction}

Let $\cal P$ be a finite classical polar space, i.e., $\cP$ arises from a vector space of finite dimension over a finite field equipped with a
non--degenerate reflexive sesquilinear form. Hence $\cP$ is a member of one of the following classes: a symplectic space ${\cal W}(2n+1,q)$, a quadric ${\cal Q}(2n,q)$, ${\cal Q}^+(2n+1,q)$, ${\cal Q}^-(2n+1,q)$ or a Hermitian variety ${\cal H}(n,q^2)$.
A projective subspace of maximal dimension contained in $\cal P$ is called a {\em generator} of $\cal P$.

For further details on finite classical polar spaces we refer to \cite{HT},\cite{S}.

	\begin{definition}
A {\bf partial spread} $\cal S$ of $\cal P$ is a set of pairwise disjoint generators. A partial spread is said to be {\bf maximal} if it is maximal with respect to set--theoretic inclusion. A partial spread $\cal S$ is called a {\bf spread} if $\cal S$ partitions the point set of $\cal P$.
	\end{definition}

If a polar space does not admit spreads, the question on the size of a maximal partial spread in such a space naturally arises. In general constructing maximal partial spreads and obtaining reasonable upper and lower bounds for the size of such partial spreads is an interesting problem. Recently maximal partial spreads of symplectic polar spaces received particular attention due to their applications in quantum information theory. In fact they correspond to so--called weakly unextendible mutually unbiased bases \cite{Grassl},\cite{KT}.
In this paper we are interested in maximal partial spreads of ${\cal H}(2n+1,q^2)$, ${\cal Q}^+(4n-1,q)$, ${\cal Q}(4n-2,q)$, $n\ge 2$, for any $q$ and of ${\cal W}(2n+1,q)$, $n\ge 2$,  when $q$ is even. In particular we show that, for $n \ge 1$, $\cH(4n-1,q^2)$ has a maximal partial spread of size $q^{2n}+1$, $\cH(4n+1,q^2)$ has a maximal partial spread of size $q^{2n+1}+1$ and, for $n \ge 2$, $\cQ^+(4n-1,q)$, $\cQ(4n-2,q)$, $\cW(4n-1,q)$, $q$ even, $\cW(4n-3,q)$, $q$ even, have a maximal partial spread of size $q^n+1$. These results are obtained by investigating particular Segre varieties $\cS_{1,n}$ ``embedded'' in polar spaces of Hermitian or hyperbolic type.


	\section{The geometric setting}

We will use the term $n$-space to denote an $n$--dimensional projective subspace of the ambient projective space. Also in the sequel we will use the following notation $\theta_{n,q} := \genfrac{[}{]}{0pt}{}{n+1}{1}_q=q^n + \cdots + q + 1$.

    \subsection{Linear representations and spreads of projective spaces}

Let $(V, k)$ be a non--degenerate formed space with associated polar space $\cP$ where $V$ is a $(d+1)$--dimensional vector space over $\GF(q^e)$ and $k$ is a sesquilinear (quadratic) form. The vector space $V$ can be considered as an $(e(d+1))$--dimensional vector space $V'$ over $\GF(q)$ via the inclusion $\GF(q) \subset \GF(q^e)$. Composition of $k$ with the trace map $Tr : z \in \GF(q^e) \mapsto \sum_{i=1}^{e} z^{q^i} \in \GF(q)$ provides a new form $k'$ on $V'$ and so we obtain a new formed space $(V',k')$. If our new formed space $(V',k')$ is non--degenerate, then it has an associated polar space $\cP'$. The isomorphism types and various conditions are presented in \cite{KL}, \cite{Gill}. Now each point in $\PG(d,q^e)$ corresponds to a $1$--dimensional vector subspace in $V$, which in turn corresponds to an $e$--dimensional vector subspace in $V'$, that is an $(e-1)$--space of $\PG(e(d + 1)-1,q)$. Extending this map from points of $\PG(d,q^e)$ to subspaces of $\PG(d,q^e)$, we obtain an injective map from subspaces of $\PG(d,q^e)$ to certain subspaces of $\PG(e(d+1)-1,q)$:
$$
 \phi: \PG(d,q^e) \rightarrow \PG(e(d+1)-1,q).
$$
The map $\phi$ is called the $\GF(q)$--{\em linear representation} of $\PG(d,q^e)$.

A {\em $t$--spread} of a projective space $\bP$ is a collection $\cS$ of mutually disjoint $t$--spaces of $\bP$ such that each point of $\bP$ is contained in an element of $\bP$. The set $\cD=\{\phi(P) \; | \;\; P \in \PG(d,q^e)\}$ is an example of $(e-1)$--spread of $\PG(e(d+1)-1,q)$, called a {\it Desarguesian spread} (see \cite{S1}, Section 25). The incidence structure whose points are the elements of $\cD$ and whose lines are the $(2e-1)$--spaces of $\PG(e(d+1)-1,q)$ joining two distinct elements of $\cD$, is isomorphic to $\PG(d,q^e)$. One immediate consequence of the definitions is that the image of the pointset of the original polar space $\cP$ is contained in the new polar space $\cP'$ (but is not necessarily equal to it).

	\subsection{Segre varieties $\cS_{1,n}$}
	
Consider the map defined by 	
$$
\sigma:\PG(1,q)\times\PG(n,q)\to \PG(2n+1,q),
$$
taking a pair of points $x=(x_1, x_2)$ of $\PG(1,q)$, $y= (y_1, \dots, y_n, y_{n+1})$ of $\PG(n,q)$ to their product $(x_1y_1, \dots, x_2y_{n+1})$. This is a special case of a wider class of maps called {\em Segre maps} \cite{HT}.
The image of $\sigma$ is an algebraic variety called the {\em Segre variety} and denoted by ${\cal S}_{1,n}$. The Segre variety ${\cal S}_{1,n}$ has two rulings, say ${\cal R}_1$ and ${\cal R}_2$, where ${\cal R}_1$ contains $\theta_{n,q}$ lines and ${\cal R}_2$ consists of $q+1$ $n$--spaces such that two subspaces in the same ruling are disjoint, and each point of ${\cal S}_{1,n}$ is contained in exactly one member of each ruling. Also, a member of ${\cal R}_1$ meets an element of ${\cal R}_2$ in exactly one point.

Notice that the set $\cR_1$ consists of all the lines of $\PG(2n+1,q)$ incident with three distinct members of $\cR_2$ and, from \cite[Theorem 25.6.1]{HT}, three mutually disjoint $n$--spaces of $\PG(2n+1,q)$ define a unique Segre variety ${\cal S}_{1,n}$. A line of $\PG(2n+1,q)$ shares with $\cS_{1,n}$ $0,1,2$ or $q+1$ points. Also, the automorphism group of ${\cal S}_{1,n}$ in $\PGL(2n+2,q)$ is a group isomorphic to $\PGL(2,q) \times \PGL(n+1,q)$ \cite[Theorem 25.5.13]{HT}. For more details on Segre varieties, see \cite{HT}.

    \subsection{$\cS_{1,n}$ and Hermitian varieties}

Let $\cH(2n+1,q^2)$ be a Hermitian variety of $\PG(2n+1,q^2)$ and let $G \simeq \PGU(2n+2,q^2)$ be the stabilizer of $\cH(2n+1,q^2)$ in $\PGL(2n+2,q^2)$. Let $g_1,g_2,g_3$ be three mutually skew generators of ${\cal H}(2n+1,q^2)$. We recall the following lemma due to J.A. Thas \cite[p. 538]{T}.

	\begin{lemma}\label{thas}
The points of $g_1$, that lie on a line of ${\cal H}(2n+1,q^2)$ intersecting $g_2$ and $g_3$, form a Hermitian variety ${\cal H}(n,q^2)$ in $g_1$. 	
	\end{lemma}

    \begin{remark}\label{defpol}
{\rm If $\perp$ denotes the unitary polarity of $\PG(2n+1,q^2)$ induced by ${\cal H}(2n+1,q^2)$ and $\perp'$ the unitary polarity of $g_1$ induced by ${\cal H}(n,q^2)$, then $\perp'$ is defined as follows. Let $P$ be a point of $g_1$ and let $\ell$ be the unique line of $\PG(2n+1,q^2)$ through $P$ intersecting $g_2$ and $g_3$, too. Then $P^{\perp'} = g_1 \cap \ell^\perp$. }
    \end{remark}

From \cite[Table 4.2.A]{KL}, the stabilizer of $g_1$ and $g_2$ in $G$ contains a subgroup $K$ isomorphic to $\PGL(n+1,q^2)$ inducing a group isomorphic to $\PGL(n+1,q^2)$ on both $g_1$ and $g_2$. If $\cS_{1,n}$ denotes the unique Segre variety containing $g_1,g_2,g_3$, then it follows that, among the $\theta_{n,q^2}$ lines of $\cR_1$, there are $|\cH(n,q^2)|$ lines that are contained in $\cH(2n+1,q^2)$ and $\theta_{n,q^2}-|\cH(n,q^2)|$ that are $(q+1)$--secant to $\cH(2n+1,q^2)$. Hence, we have that $q+1$ members of $\cR_2$ are generators of $\cH(2n+1,q^2)$ and the remaining $q^2-q$ elements of $\cR_2$ meet $\cH(2n+1,q^2)$ in a Hermitian variety $\cH(n,q^2)$.

If $\cH_{1,n} := \cS_{1,n} \cap \cH(2n+1,q^2)$, it turns out that
$$
|\cH_{1,n}| = (q+1) \theta_{n,q^2} + (q^2-q) |\cH(n,q^2)|.
$$
We will refer to $\cH_{1,n}$ as {\em Hermitian Segre variety} of $\cH(2n+1,q^2)$.

Let $H$ be the subgroup of $K$ fixing $\cH_{1,n}$. Notice that $|H| \le |\PGU(n+1,q^2)|$. From \cite[Corollary 12]{KMS} the number of generators of ${\cal H}(2n+1,q^2)$ disjoint from $g_1$ and $g_2$ equals
$$
q^{\frac{n(n+1)}{2}}\prod_{i=1}^{n+1}(q^i+(-1)^i).
$$
On the other hand, the number of Hermitian varieties ${\cal H}(n,q^2)$ in $g_1$ equals the index of $\PGU(n+1,q^2)$ in $\PGL(n+1,q^2)$, that is
$$
\vert\PGL(n+1,q^2):\PGU(n+1,q^2)\vert = q^{\frac{n(n+1)}{2}}\prod_{i=2}^{n+1}(q^i+(-1)^i).
$$
Therefore we may conclude that $H \simeq \PGU(n+1,q^2)$.	
We have proved the following proposition which shows that the converse of Thas's lemma holds true.
	
	\begin{proposition}\label{corr}
There exists a one to one correspondence between the set of Hermitian Segre varieties $\cH_{1,n}$ of ${\cal H}(2n+1,q^2)$ containing $g_1$ and $g_2$ and the set of Hermitian varieties ${\cal H}(n,q^2)$ of $g_1$.  	
	\end{proposition}

\begin{remark}
{\rm From \cite[Table 4.4.A]{KL} the stabilizer of $\cH_{1,n}$ in $G$ is isomorphic to $\PGU(2,q^2) \times \PGU(n+1,q^2)$. }
\end{remark}

We will need the following lemma.

    \begin{lemma}\label{hermitian}
Let $\cS_1$ and $\cS_2$ be two distinct Hermitian Segre varieties of $\cH(2n+1,q^2)$ containing $g_1$ and $g_2$. Let $\cH_i$ be the Hermitian variety of $g_1$ determined by $\cS_i$ and let $\perp_i$ be the unitary polarity of $g_1$ induced by $\cH_i$, $i = 1,2$. Then $| \cS_1 \cap \cS_2 | = 2 \theta_{n,q^2} + x_1 (q^2-1) + x_2 (q-1)$, where $x_1 = | \{ P \in \cH_1 \cap \cH_2 \;\; | \;\; P^{\perp_1} = P^{\perp_2} \} |$ and $x_2 = | \{ P \in g_1 \setminus (\cH_1 \cap \cH_2) \;\; | \;\; P^{\perp_1} = P^{\perp_2} \} |$.
    \end{lemma}
    \begin{proof}
Let $\cS'_i$ be the Segre variety of the ambient projective space such that $\cS_i = \cS_i' \cap \cH(2n+1,q^2)$, $i = 1,2$. Notice that if $P \notin g_1 \cup g_2$, then there exists a unique line $\ell$ through $P$ meeting both $g_1$ and $g_2$. Therefore if $P \in \cS_1' \cap \cS_2'$, then $\ell \subset \cS_1' \cap \cS_2'$. Let $P_i = \ell \cap g_i$, $i = 1,2$ and assume that $P \in \cS_1 \cap \cS_2$. Then $P_1^{\perp_1} = \ell^\perp \cap g_1 = P^\perp \cap g_1 = P_1^{\perp_2}$. On the other hand, assume that there exists a point $P_1 \in g_1$ such that $P_1^{\perp_1} = P_1^{\perp_2}$. Let $\ell'_i$ be the unique line of $\cS'_i$, $i = 1,2$, passing through $P_1$ and having a point in common with $g_2$. Then $(P_1^{\perp_i})^\perp \cap g_2$ coincides with the point $g_2 \cap \ell'_i$, $i = 1,2$. Therefore we have $g_2 \cap \ell'_1 = g_2 \cap \ell'_2$ and hence $\ell'_1 = \ell'_2$. The result now follows from the fact that if $\ell \subset \cS_1' \cap \cS_2'$ is a line meeting both $g_1$ and $g_2$, then $| \ell \cap \cH(2n+1,q^2) | = q^2+1$ if and only if $\ell \cap g_1 \in \cH_1 \cap \cH_2$.
    \end{proof}

    \subsection{$\cS_{1,n}$ and hyperbolic quadrics}

Let $\cQ^+(4n-1,q)$ be a hyperbolic quadric of $\PG(4n-1,q)$ and let $G \simeq \PGO^+(4n,q)$ be the stabilizer of $\cQ^+(4n-1,q)$ in $\PGL(4n,q)$. Let $g_1,g_2,g_3$ be three mutually skew generators of ${\cal Q}^+(4n-1,q)$. We recall the following lemma due to A. Klein, K. Metsch and L. Storme \cite[Theorem 6]{KMS}.

	\begin{lemma}\label{thas1}
The lines of $g_1$ lying in a solid of $\cQ^+(4n-1,q)$ that intersects $g_2$, $g_3$ in a line, are those of a symplectic space $\cW(2n-1,q)$ in $g_1$.
	\end{lemma}

From \cite[Table 4.2.A]{KL}, the stabilizer of $g_1$ and $g_2$ in $G$ contains a subgroup $K$ isomorphic to $\PGL(2n,q)$ fixing both  $g_1$ and $g_2$. If $\cS_{1,2n-1}$ denotes the unique Segre variety containing $g_1,g_2,g_3$, then it follows that the $\theta_{2n-1,q}$ lines of $\cR_1$ are contained in $\cQ^+(4n-1,q)$. Hence, we have that the $q+1$ members of $\cR_2$ are generators of $\cQ^+(4n-1,q)$. In particular $\cS_{1,2n-1} \subset \cQ^+(4n-1,q)$ and, in this case, we will refer to $\cS_{1,2n-1}$ as {\em hyperbolic Segre variety} of $\cQ^+(4n-1,q)$.

Let $H$ be the subgroup of $K$ fixing a hyperbolic Segre variety of $\cQ^+(4n-1,q)$. Notice that $|H| \le |\PSp(2n,q)|$. From \cite[Corollary 5]{KMS} the number of generators of ${\cal Q}^+(4n-1,q)$ disjoint from $g_1$ and $g_2$ equals
$$
q^{n(n-1)}\prod_{i=1}^{n}(q^{2i-1}-1).
$$
On the other hand, the number of symplectic polar space ${\cal W}(2n-1,q)$ of $g_1$ is equal to
$$
\vert\PGL(2n,q):\PSp(2n,q)\vert = q^{n(n-1)}\prod_{i=2}^{n}(q^{2i-1}-1).
$$
Therefore we may conclude that $H \simeq \PSp(2n,q)$.	
We have proved the following proposition which shows that the converse of Lemma \ref{thas1} holds true.
	
	\begin{proposition}\label{corr1}
There exists a one to one correspondence between the set of hyperbolic Segre varieties of ${\cal Q}^+(4n-1,q)$ containing $g_1$ and $g_2$ and the set of symplectic spaces ${\cal W}(2n-1,q)$ of $g_1$.  	
	\end{proposition}

\begin{remark}\label{grupposh}
{\rm From \cite[Table 4.4.A]{KL} the stabilizer of a hyperbolic Segre variety in $G$ is isomorphic to $\PSp(2,q) \times \PSp(2n,q)$. }
\end{remark}

Arguing as in Lemma \ref{hermitian}, we obtain the following.

    \begin{lemma}\label{hyperbolic}
Let $\cS_1$ and $\cS_2$ be two distinct hyperbolic Segre varieties of $\cQ^+(4n-1,q)$ containing $g_1$ and $g_2$. Let $\cW_i$ be the symplectic polar space of $g_1$ determined by $\cS_i$ and let $\perp_i$ be the symplectic polarity of $g_1$ induced by $\cW_i$, $i = 1,2$. Then $| \cS_1 \cap \cS_2 | = 2 \theta_{2n-1,q} + x (q-1)$, where $x = | \{ P \in g_1 \;\; | \;\; P^{\perp_1} = P^{\perp_2} \} |$.
    \end{lemma}

	\section{Hermitian polar spaces}

    \subsection{$\cH(4n-1,q^2)$, $n \ge 1$}

Let $\cH_1$ and $\cH_2$ be the two distinct Hermitian varieties of $\PG(4n-1,q^2)$ having the following homogeneous equations
$$
 f_1 : X_1 X_{2n+1}^q + \cdots + X_{2n} X_{4n}^q + X_1^q X_{2n+1} + \cdots + X_{2n}^q X_{4n} = 0,
$$
$$
 f_2 : X_1 X_{2n+1}^q + \cdots + X_{2n} X_{4n}^q + \omega^{q-1} ( X_1^q X_{2n+1} + \cdots + X_{2n}^q X_{4n} ) = 0,
$$
respectively, where $\omega$ is a primitive element of $\GF(q^2)$. Then the Hermitian pencil $\cF$ defined by $\cH_1$ and $\cH_2$ is the set of all Hermitian varieties with equations $a f_1 + b f_2 = 0$, as $a$ and $b$ vary over the subfield $\GF(q)$, not both zero. Note that there are $q+1$ distinct Hermitian varieties in the pencil $\cF$, none of which is degenerate. The set $\cX = \cH_1 \cap \cH_2$ is called the base locus of $\cF$. Since the Hermitian varieties of a pencil cover all the points of $\PG(4n-1,q^2)$, a counting argument shows that
$$
 |\cX| = \frac{(q^{4n-2}+1)(q^{4n}-1)}{q^2-1}
$$
and any two distinct varieties in $\cF$ intersect precisely in $\cX$.
In particular $\cX$ is a variety defined by the following equation:
$$
  X_1 X_{2n+1}^q + \cdots + X_{2n} X_{4n}^q = 0.
$$
Straightforward computations show that\,$\cX$\,contains the following two\,$(2n-1)$--dimensional projective spaces:
$$
 \Sigma : X_1 = \cdots = X_{2n} = 0,  \Sigma' : X_{2n+1} = \cdots = X_{4n} = 0.
$$
Also, through a point $P$ of $\Sigma$ (resp. $\Sigma'$) there pass $\theta_{2n-2,q^2}$ lines entirely contained in $\cX$ and these lines are contained in a generator of $\cH(4n-1,q^2)$ meeting $\Sigma$ (resp. $\Sigma'$) exactly in $P$.

Let $\Pi_{r-1}$ be an $(r-1)$--dimensional projective space of $\Sigma$, $1 \le r \le 2n-1$, and let $\Pi_{r-1}^\perp$ be the polar space of $\Pi_{r-1}$ with respect to the unitary polarity $\perp$ of $\cH_1$ (or, equivalently, $\cH_2$). The intersection of $\Pi_{r-1}^\perp$ and $\Sigma'$ is a $(2n-r-1)$--dimensional projective space, say $\Pi'_{2n-r-1}$. Note that $\langle \Pi_{r-1}, \Pi'_{2n-r-1} \rangle$ is a generator of $\cH_1$ contained in $\cX$. In particular, an $(n-1)$--dimensional subspace of $\Sigma$ corresponds to a uniquely determined $(n-1)$--dimensional subspace of $\Sigma'$.

Let $\cal D$ be an $(n-1)$--spread of $\Sigma$. From the above discussion, $\cal D$ defines an $(n-1)$--spread, say ${\cal D}'$ of $\Sigma'$. Corresponding elements of $\cal D$ and $\cal D'$ determine a generator of any member of $\cal F$ and hence $\cal D$ (${\cal D'}$) produces a partial spread ${\cal S}$ of $\cH_1$ of size $\vert{\cal D}\vert =q^{2n}+1$.

In the next result we prove that the partial spread $\cal S$ is maximal.

	\begin{theorem}\label{spr}
Any $(n-1)$--spread $\cal D$ of $\PG(2n-1,q^2)$ gives rise to a maximal partial spread $\cal S$ of ${\cal H}(4n-1,q^2)$ of size $q^{2n}+1$.
	\end{theorem}
	\begin{proof}
Let $\bar{\cal S}$ denote the pointset of $\cH_1$ covered by $\cal S$. 	
Let $\cT$ be a generator of $\cH_1$ disjoint from $\Sigma$ and $\Sigma'$ and let $\cH_{1,2n-1}$ be the Hermitian Segre variety obtained by intersecting the unique Segre variety ${\cS}_{1,2n-1}$ defined by $\Sigma,\Sigma'$ and $\cT$ with $\cH_1$. Let ${\cal H}_\Sigma$ and ${\cal H}_{\Sigma '}$ denote the Hermitian varieties of $\Sigma$ and $\Sigma'$, respectively, arising from Thas's lemma. Let $d$ be a member of $\cal D$. Then $d^\perp\cap\Sigma'$ defines a unique member $d'$ of ${\cal D}'$. Here $\perp$ denotes the unitary polarity induced by $\cH_1$. Let $L_d \subset \cR_1$ be the set of lines of $\cS_{1,2n-1}$ meeting $d$. The set $L_d$ determines an $(n-1)$--space of $\Sigma'$, say $d''$. From Remark \eqref{defpol}, it turns out that $d'$ and $d''$ are conjugate with respect to the polarity of $\Sigma'$ induced by $\cH_{\Sigma'}$.

Since members of ${\cal D}'$ are pairwise disjoint and $\vert {\cal H}_{\Sigma'}\vert =(q^{2n+2}-1)(q^{2n+1}+1)/(q^2-1)$, it follows that there exists at least one member $X$ of ${\cal D}'$ meeting ${\cal H}_{\Sigma'}$ in a degenerate Hermitian variety.  
With respect to the unitary polarity of $\Sigma'$ induced by $\cH_{\Sigma'}$ the conjugate of $X$ meets $X$ in at least a point, say $P$, of $\cH_{\Sigma'}$. Let $\ell$ be the unique line of ${\cS}_{1,2n-1}$ through $P$.
Then, on one hand $\ell$ meets $\cT$ in a point $Q$. On the other hand, $\ell$ is contained $\langle d,d'\rangle$. It follows that $\cT$ has a non--empty intersection with $\bar{\cal S}$. 	
	\end{proof}
	
	\begin{remark}
{\rm The automorphism group of the maximal partial spread $\cal S$ in $\PGU(4n,q^2)$ contains the automorphism group of $\cal D$ as a subgroup of $\PGL(2n,q^2)$.}	
	\end{remark}
	
	\begin{remark}\label{Luyckx}
{\rm In \cite{ACE} the authors observed that the $q^{m+1}+1$ elements of a spread of ${\cal W}(2m+1,q)$ embedded in ${\cal H}(2m+1,q^2)$ extend to pairwise disjoint generators of ${\cal H}(2m+1,q^2)$. Maximality of partial spreads of $\cH(2m+1,q^2)$ constructed in this way was shown for $m=1,2$ in \cite{ACE} and for all even $m$ in \cite{L}. When $m=2n-1$, $m>1$, it is not immediately clear whether this partial spread is maximal or not. From Theorem \ref{spr} it can be deduced that if the spread of $\cW(4n-1,q)$ arises from a spread of $\cW(2n-1,q^2)$ in the $\GF(q)$--linear representation, then the ``extended'' partial spread of $\cH(4n-1,q^2)$ is maximal. Indeed, if $\cL$ is a Desarguesian line--spread of a ${\cal W}(4n-1,q)$ embedded in ${\cal H}(4n-1,q^2)$, then we may assume that $\Sigma$ and $\Sigma'$ play the role of director spaces of $\cL$. Extending the elements of a spread $\cS'$ of $\cW(4n-1,q)$, we get an $(n-1)$--spread of $\Sigma$ ($\Sigma'$) if and only if $\cL$ induces a line--spread on each of the members of $\cS'$. Under this assumption, the partial spread of $\cH(4n-1,q^2)$, obtained by extending the elements of $\cS'$, coincides with the construction given above. }	
	\end{remark}

    \subsection{$\cH(4n+1,q^2)$, $n \ge 1$}

As already observed in Remark \ref{Luyckx}, the $q^{2n+1}+1$ elements of a spread of ${\cal W}(4n+1,q)$ embedded in ${\cal H}(4n+1,q^2)$ extend to pairwise disjoint generators of ${\cal H}(4n+1,q^2)$ forming a maximal partial spread of $\cH(4n+1,q^2)$ as shown in \cite{ACE} for $n=1$ and in \cite{L} for $n \ge 1$. On the other hand, it is known that the size of a partial spread in $\cH(4n+1,q^2)$ is at most $q^{2n+1}+1$, see \cite{BM}, \cite{V1}, \cite{V2}. In what follows we prove that $\cH(4n+1,q^2)$ has a maximal partial spread attaining the upper bound. We do not know if such a partial spread arises from a spread of a $\cW(4n+1,q)$ embedded in $\cH(4n+1,q^2)$.
\medskip

From \cite{CS} we recall the following facts. Consider the projective space $\PG(2n,q^2)$ in its field--model representation: the points of $\PG(2n,q^2)$ are the nonzero elements of $\GF(q^{4n+2})$ and two elements $x,y \in \GF(q^{4n+2})$  define the same point of $\PG(2n,q^2)$ if and only if $x/y \in \GF(q^2)$.
The point of $\PG(2n,q^2)$ represented by $x\in\GF(q^{4n+2})\setminus\{0\}$ will be denoted by $(x)$. The hyperplanes of $\PG(2n,q^2)$ are the sets $[u]:=\{(x) \in \GF(q^{4n+2})\pmod{\GF(q^2)} \;\; | \;\; Tr(ux)=0\}$, where $Tr$ denotes the usual trace map from $\GF(q^{4n+2})$ to $\GF(q^2)$, $u\in  \GF(q^{4n+2})\setminus\{0\}$. In this setting the map $\sigma : (x) \mapsto (\omega x)$, where $\omega$ is a primitive element of $\GF(q^{4n+2})$, is a Singer cycle of $\PG(2n,q^2)$.

Let $a\in\GF(q^{4n+2})\setminus\{0\}$ such that $a^{q^{2n+1}-1}\in\GF(q^2)$. Then
$$
\Pi_a:
\begin{cases}
(x)\mapsto [ax^{q^{2n+1}}], & x\in\GF(q^{4n+2})\setminus\{0\}\\
[u]\mapsto (u^{q^{2n+1}}/a), & u\in \GF(q^{4n+2})\setminus\{0\}
\end{cases}
$$
is a unitary polarity of $\PG(2n,q^2)$. In particular, varying $a$, we get a set consisting of $\theta_{2n,q}$ Hermitian varieties, such that the intersection of any $2n$ of them is a $(q^{2n+1}+1)/(q+1)$--cap (i.e., a set of points such that no three of them are collinear) and there is a partition of $\PG(2n,q^2)$ into such caps which is invariant under a Singer cyclic subgroup of order $(q^{2n+1}+1)/(q+1)$.

	\begin{proposition} \label{fpf}
Let $a_i \in\GF(q^{4n+2})\setminus\{0\}$ such that $a_i^{q^{2n+1}-1}\in\GF(q^2)$, $i = 1,2$ and $a_1 \ne a_2$. Then $\Pi_{a_1} \Pi _{a_2}$ is a fixed point free projectivity of $\PG(2n,q^2)$.	
	\end{proposition}
	\begin{proof}
Since $a_i = \omega^{k_i (q^{2n+1}+1)/(q+1)}$, $k_1 \ne k_2$, \cite[Theorem 2.2]{CS}, it follows that the projectivity $\Pi_{a_1} \Pi _{a_2}$ sends $(x)$ to $(\omega^{(k_2 q^{2n+1}-k_1)(q^{2n+1}+1)/(q+1)} x)$.
	\end{proof}

We are ready to give the aforementioned construction.

	\begin{theorem} \label{spr1}
$\cH(4n+1,q^2)$ has a maximal partial spread of size $q^{2n+1}+1$ admitting a cyclic group of order $\theta_{2n,q^2}$ as an automorphism group.
	\end{theorem}
	\begin{proof}
From Proposition \ref{corr}, Lemma \ref{hermitian} and Proposition \ref{fpf}, we find $\theta_{2n,q}$ Hermitian Segre varieties containing two fixed generators $g_1, g_2$ of $\cH(4n+1,q^2)$ and pairwise sharing exactly $g_1 \cup g_2$. Hence we get a set $\cS$ of $\theta_{2n,q} (q-1) + 2 = q^{2n+1}+1$ mutually disjoint generators. Of course $\cS$ is maximal and by construction it is left invariant by a cyclic group of order $\theta_{2n,q^2}$ corresponding to a Singer cyclic group of $g_1$.
	\end{proof}

	\begin{remark}
{\rm If $\cD$ is the set of extended elements of a Desarguesian spread of a $\cW(4n+1,q)$ embedded in $\cH(4n+1,q^2)$, then it is easily seen that the $q+1$ generators contained in the Hermitian Segre variety determined by any three distinct elements of $\cD$ are contained in $\cD$. It is therefore plausible that the maximal partial spread constructed in Theorem \ref{spr1} is equivalent to $\cD$. }
	\end{remark}

	\section{Orthogonal and symplectic polar spaces}	

Let $\cQ^+(2n+1,q)$ be a hyperbolic quadric of $\PG(2n+1,q)$. The set of all generators of $\cQ^+(2n+1,q)$, that are $n$--spaces, is divided in two distinct subsets of the same size, called {\em systems of generators} and denoted by $\cM_1$ and $\cM_2$, respectively. Let $A$ and $A'$ two distinct generators of $\cQ^+(2n+1,q)$. Then their possible intersections are projective spaces of dimension
$$
\left\{
\begin{array}{ccccccc}
0, & 2, & 4, & \ldots, & n-2 & \mbox{    if    }  & A, A' \in \cM_i, i=1,2 \\
-1, & 1, & 3, & \ldots, & n-1 & \mbox{    if    }  & A \in \cM_i, A' \in \cM_j, i,j \in \{ 1,2 \}, i \ne j
\end{array}
\right.
$$
if $n$ is even or
$$
\left\{
\begin{array}{ccccccc}
0, & 2, & 4, & \ldots, & n-1 & \mbox{    if    }  & A \in \cM_i, A' \in \cM_j, i,j \in \{ 1,2 \}, i \ne j \\
-1, & 1, & 3, & \ldots, & n-2 & \mbox{    if    }  & A, A' \in \cM_i, i=1,2
\end{array}
\right.
$$
if $n$ is odd.

It follows that, if $n$ is even, a maximal partial spread of $\cQ^+(2n+1,q)$ has size two. Therefore we will assume that $n=2m-1$ and in this case all the elements of a partial spread of $\cQ^+(4m-1,q)$ belong to the same system of generators.
\begin{remark} \label{para}
{\rm Let $\cQ(4m-2,q)$ be a parabolic quadric obtained by intersecting $\cQ^+(4m-1,q)$ with a non--degenerate hyperplane section. Then a (maximal) partial spread of $\cQ^+(4m-1,q)$ induces a (maximal) partial spread of $\cQ(4m-2,q)$. Actually the converse also holds true.}
\end{remark}

Let $\cQ^+(3,q^m)$ be a hyperbolic quadric of $\PG(3,q^m)$, $m \ge 2$. In this special case $\cQ^+(3,q^m)$ coincides with $\cS_{1,1}$. Hence, both systems of generators, $\cR_1$ and $\cR_2$, consist of $q^m+1$ mutually disjoint lines each, and are called {\em reguli}, see \cite{H}.
Under the $\GF(q)$--linear representation of $\PG(3,q^m)$ the points of $\cQ^+(3,q^m)$ are mapped to $(m-1)$--spaces of a hyperbolic quadric $\cQ^+(4m-1,q)$ and a line of $\cQ^+(3,q^m)$ is mapped to a generator of $\cQ^+(4m-1,q)$. It follows that $\phi(\cR_1) = \{ \phi(\ell) \;\; | \;\; \ell \in \cR_1 \}$ is a partial spread of $\cQ^+(4m-1,q)$ of size $q^m+1$. We denote by $\cQ$ the points of $\cQ^+(4m-1,q)$ covered by members of $\phi(\cQ^+(3,q^m))$.

    \begin{theorem} \label{regulus}
$\phi(\cR_i)$ is a maximal partial spread of $\cQ^+(4m-1,q)$.
	\end{theorem}
    \begin{proof}
Let $\ell_i \in \cR_i$, $i=1,2$. Then $\phi(\ell_1) \cap \phi(\ell_2)$ is an $(m-1)$--space.
\medskip

\fbox{$m$ odd}

Since $m$ is odd, we have that $\phi(\ell_1)$ and $\phi(\ell_2)$ belong to distinct systems of generators of $\cQ^+(4m-1,q)$. Therefore, we may assume that $\phi(\ell_i) \in \cM_i$. This means that $\phi(\cR_i) \subset \cM_i$. The result follows.
\medskip

\fbox{$m$ even}

In this case we may assume that both $\phi(\cR_1)$, $\phi(\cR_2)$ are contained in $\cM_1$. Let $g \in \cM_2$. Then $| g \cap \cQ | \ge q^m+1$. If $g \cap \cQ$ contains a line, then every $(2m-2)$--space contained in $g$ has at least one point in common with $\cQ$. If $g \cap \cQ$ does not contain any line, then every element of $\phi(\cR_1)$ (and $\phi(\cR_2)$) has exactly one point in common with $g$. Let $P_1, P_2, P_3$ be three distinct points of $g \cap \cQ$. Let $g_i \in \phi(\cR_1)$ such that $g_i \cap g = \{ P_i \}$, $i = 1,2,3$. Then, since $g_1, g_2, g_3$ are three distinct mutually disjoint generators of $\cQ^+(4m-1,q)$, there exists a unique hyperbolic Segre variety $\cS_{1,2m-1}$ of $\cQ^+(4m-1,q)$ containing $g_1, g_2, g_3$. First of all notice that the $q+1$ $(2m-1)$--spaces contained in $\cS_{1,2m-1}$ are also contained in $\phi(\cR_1)$. Indeed, if $A_i = \ell_2 \cap \phi^{-1}(g_i)$, $i = 1,2,3$, then there exists a unique subline of order $q$, say $s$, of $\ell_2$ containing $A_1, A_2, A_3$. Therefore $\{ \phi(\ell) \;\; | \;\; \ell \in \cR_1, \ell \cap \ell_2 \in s \}$ is the set of $q+1$ $(2m-1)$--spaces contained in $\cS_{1,2m-1}$. Let $\pi$ be the plane $\langle P_1, P_2, P_3 \rangle$ and let $r_i$ be a line of $g_i$ passing through $P_i$, $i = 1,2,3$. Then $r_1, r_2, r_3$ are three mutually disjoint lines and the set of points covered by the lines incident to $r_1, r_2, r_3$ is a three--dimensional hyperbolic quadric contained in $\cS_{1,2m-1}$. It follows that $\pi \cap \cQ = \pi \cap \cS_{1,2m-1}$ is a conic, say $C$. Let $P_4$ be a point of $(g \cap \cQ) \setminus C$. By considering the planes $\langle P, Q, P_4 \rangle$, where $P, Q$ are distinct points of $C$ and using again the previous argument we obtain that the solid $\langle P_1,P_2,P_3,P_4\rangle\subset g$ meets $\cQ$ in a three--dimensional elliptic quadric. Hence, it turns out that every $(2m-2)$--space contained in $g$ has at least a point in common with $\cQ$. The result follows from the fact that through every $(2m-2)$--space contained in $\cQ^+(4m-1,q)$ there passes exactly one generator of $\cM_1$ and one of $\cM_2$.
    \end{proof}

\begin{remark}
{\rm The idea of considering the image of a $\cQ^+(3,q^m)$ under the $\GF(q)$--linear representation to construct maximal partial spread of polar spaces was first pointed out by M. Grassl during the conference ALCOMA '15. In particular, using completely different techniques, he presented a proof of the result of Corollary \ref{Grassl}.}
\end{remark}

Now, we focus on the even characteristic case.
\begin{proposition}
Assume that $q$ is even. If $\cQ^+(4m-1,q)$ has a maximal partial spread of size $x$ then
\begin{itemize}
\item $\cW(4m-1,q)$ has a maximal partial spread of size $x$,
\item $\cW(4m-3,q)$ has a maximal partial spread of size $x$.
\end{itemize}
\end{proposition}
\begin{proof}
Let $\cS \subset \cM_1$ be a maximal partial spread of $\cQ^+(4m-1,q)$ and let $\bar{\cS}$ be the set of points covered by the members $\cS$. Let $X$ be an element of $\cM_1 \setminus \cS$. Since $\cS$ is maximal, there exists $Y \in \cS$ such that $|X \cap Y| \ne 0$. It follows that $X$ contains at least a line of $\bar{\cS}$. Hence, as through a $(2m-2)$--space contained in $\cQ^+(4m-1,q)$ there passes a generator of $\cM_1$, we have that every $(2m-2)$--space contained in $\cQ^+(4m-1,q)$ has at least one point of $\bar{\cS}$.

As $q$ is even, there exists a unique symplectic polar space $\cW(4m-1,q)$ of $\PG(4m-1,q)$ such that all the lines tangent to or contained in $\cQ^+(4m-1,q)$ are lines of $\cW(4m-1,q)$. In other words, the orthogonal polarity defined by $\cQ^+(4m-1,q)$ polarizes to the symplectic polarity induced by $\cW(4m-1,q)$. In this setting, a generator of $\cW(4m-1,q)$ that does not belong to $\cM_1 \cup \cM_2$, meets $\cQ^+(4m-1,q)$ in a $(2m-2)$--space. Furthermore, through a $(2m-2)$--space contained in $\cQ^+(4m-1,q)$ there pass $q-1$ generators of $\cW(4m-1,q)$ not contained in $\cM_1 \cup \cM_2$. It turns out that every generator of $\cW(4m-1,q)$ contains at least a point of $\bar{\cS}$ and hence $\cS$ is a maximal partial spread of $\cW(4m-1,q)$.

The second assertion follows from Remark \ref{para} and the fact that, for $q$ even, $\cQ(4m-2,q)$ is isomorphic to $\cW(4m-3,q)$ \cite[Theorem 4]{T}.
\end{proof}

In the following result, with similar arguments to that used in the previous section for Hermitian polar spaces, we give a construction of maximal partial spreads of $\cQ^+(4m-1,q)$, $q$ even.

    \begin{theorem}\label{spr2}
Any $(m-1)$--spread of $\PG(2m-1,q)$, $q$ even, gives rise to a maximal partial spread of ${\cal Q}^+(4m-1,q)$ of size $q^{m}+1$.
	\end{theorem}
    \begin{proof}
Let $\Sigma$ and $\Sigma'$ be two disjoint generators of $\cQ^+(4m-1,q)$. Let $\Pi_{r-1}$ be a $(r-1)$--dimensional projective space of $\Sigma$, $1 \le r \le 2m-1$, and let $\Pi_{r-1}^\perp$ be the polar space of $\Pi_{r-1}$. Here $\perp$ denotes the orthogonal polarity induced by $\cQ^+(4m-1,q)$. The intersection of $\Pi_{r-1}^\perp$ and $\Sigma'$ is a $(2m-r-1)$--dimensional projective space, say $\Pi'_{2m-r-1}$. Note that $\langle \Pi_{r-1}, \Pi'_{2n-r-1} \rangle$ is a generator of $\cQ^+(4m-1,q)$. In particular, an $(m-1)$--dimensional subspace of $\Sigma$ corresponds to a uniquely determined $(m-1)$--dimensional subspace of $\Sigma'$.

Let $\cal D$ be an $(m-1)$--spread of $\Sigma$. From the above discussion, $\cal D$ defines an $(m-1)$--spread, say ${\cal D}'$ of $\Sigma'$. Corresponding elements of $\cal D$ and $\cal D'$ determine a generator of $\cQ^+(4m-1,q)$ and hence $\cal D$ (${\cal D'}$) produces a partial spread $\cS$ of $\cQ^+(4m-1,q)$ of size $\vert{\cal D}\vert =q^{m}+1$.
Let $\bar{\cal S}$ denote the pointset of $\cQ^+(4m-1,q)$ covered by members of $\cal S$. 	
Let $\cT$ be a generator of $\cQ^+(4m-1,q)$ disjoint from $\Sigma$ and $\Sigma'$ and let $\cS_{1,2m-1}$ be the hyperbolic Segre variety of $\cQ^+(4m-1,q)$ containing $\Sigma,\Sigma'$ and $\cT$. Let ${\cal W}_\Sigma$ and ${\cal W}_{\Sigma'}$ denote the symplectic space of $\Sigma$ and $\Sigma'$, respectively, arising from Lemma \ref{thas1}. Let $d$ be a member of $\cal D$. Then $d^\perp\cap\Sigma'$ defines a unique member $d'$ of ${\cal D}'$. Let $L_d\subset \cR_1$ be the set of lines of $\cS_{1,2m-1}$ meeting $d$. The set $L_d$ determines an $(m-1)$--space of $\Sigma'$, say $d''$. It turns out that $d'$ and $d''$ are conjugate with respect to the polarity of $\Sigma'$ induced by $\cW_{\Sigma'}$.

We claim that there exists at least one member $X$ of ${\cal D}'$ meeting ${\cal W}_{\Sigma'}$ in a degenerate symplectic space. If $m$ is odd, the assertion is trivial. If $m$ is even, assume, by way of contradiction, that each of the $q^m+1$ elements of $\cD'$ meets $\cW_{\Sigma'}$ in a $\cW(m-1,q)$. Consider a hyperbolic quadric $\cQ^+(2m-1,q)$ of $\Sigma'$ such that the orthogonal polarity defined by $\cQ^+(2m-1,q)$ polarizes to the symplectic polarity induced by $\cW_{\Sigma'}$. Then every element of $\cD'$ meets $\cQ^+(2m-1,q)$ either in a $\cQ^+(m-1,q)$ or in a $\cQ^-(m-1,q)$. Let $y_1$ and $y_2$ be the number of elements of $\cD'$ meeting $\cQ^+(2m-1,q)$ in a $\cQ^+(m-1,q)$ and in a $\cQ^-(m-1,q)$, respectively. Then $y_1 + y_2 = q^m+1$, $|\cQ^-(m-1,q)| y_1 + |\cQ^+(m-1,q)| y_2 = |\cQ^+(2m-1,q)|$ imply that $y_2 = (q^{m/2}+1)^2/2$, a contradiction, since $q$ is even. So there exists an element $X$ of $\cD'$ meeting $\cW_{\Sigma'}$ in a degenerate symplectic space.

With respect to the symplectic polarity of $\Sigma'$ induced by $\cW_{\Sigma'}$ the conjugate of $X$ meets $X$ in at least a point of $\cW_{\Sigma'}$. It follows that $\cT$ has a non--empty intersection with $\bar{\cal S}$.
    \end{proof}

    \begin{corollary}
$\cQ(4m-2)$ has a maximal partial spread of size $q^m+1$.
    \end{corollary}

    \begin{corollary} \label{Grassl}
$\cW(4m-1,q)$, $q$ even, has a maximal partial spread of size $q^m+1$.
    \end{corollary}

    \begin{corollary}
$\cW(4m-3,q)$, $q$ even, has a maximal partial spread of size $q^m+1$.
    \end{corollary}

    \begin{remark}
{\rm The result of Theorem \ref{spr2} cannot be extended to the odd characteristic case. Indeed, consider in $\PG(3,3)$ a Desarguesian line--spread $\cD$ such that $\cD$ contains 4 lines of a regulus $\cR$ of a fixed symplectic space $\cW(3,3)$. In particular $\cW(3,3)$ can be chosen in such a way that none of the lines of the opposite regulus of $\cR$, say $\cR'$, is a line of $\cW(3,3)$. If $\cD'$ denotes the Hall spread $(\cD \setminus \cR) \cup \cR'$, then each of the 10 elements of $\cD'$ is non--degenerate with respect to $\cW(3,3)$. Magma computations show that the partial spread $\cS$ of $\cQ^+(7,3)$ constructed from $\cD'$ (as in Theorem \ref{spr2}) is not maximal; indeed there exist 34 generators disjoint from all the members of $\cS$. Furthermore, among the 34 disjoint generators, it is possible to find a subset of 18 generators that, together with the 10 members of $\cS$, give rise to the unique spread of $\cQ^+(7,3)$, admitting $\PSp(6,2)$ as an automorphism group. }
    \end{remark}

	\subsection{The triality quadric}
		
In this section we specialize to the case of the triality quadric ${\cal Q}^+(7,q)$. We will denote by $\perp$ the polarity of $\PG(7,q)$ induced by ${\cal Q}^+(7,q)$.

Let us denote by ${\cal M}_1$ and ${\cal M}_2$ the two families of generators of ${\cal Q}^+(7,q)$, the set of lines of ${\cal Q}^+(7,q)$ by $\cal L$ and the set of points of ${\cal Q}^+(7,q)$ by $\cal P$; then the rank $4$ incidence geometry $\Omega = (\cP, \cL, \cM_1, \cM_2)$ can be attached to ${\cal Q}^+(7,q)$ as follows. An element $G_1 \in {\cal M}_1$ is said to be incident with an element $G_2 \in {\cal M}_2$ if and only if the intersection $G_1 \cap G_2$ is a plane of ${\cal Q}^+(7,q)$. Incidence between other elements is symmetrized containment. Every permutation of the set $\{{\cal P}, {\cal M}_1, {\cal M}_2\}$ defines a geometry $\Omega'$ isomorphic to $\Omega$ and hence the automorphism groups of $\Omega$ and $\Omega'$ are isomorphic.

A {\em triality} of the geometry $\Omega$ is a map $\tau$:
$$
\tau:{\cal L}\to {\cal L}, {\cal P}\to {\cal M}_1, {\cal M}_1\to {\cal M}_2, {\cal M}_2\to {\cal P}
$$
preserving the incidence in $\Omega$ and such that $\tau^3$ is the identity.

The triality image of a maximal partial spread $\cS \subset \cM_2$ of ${\cal Q}^+(7,q)$ is a maximal partial ovoid of ${\cal Q}^+(7,q)$, i.e. a set of mutually non--collinear points of $\cQ^+(7,q)$ that is maximal with respect to set--theoretic inclusion.
\medskip

In the case of ${\cal Q}^+(7,q)$ we can extend Theorem \ref{spr2} to the case $q$ odd as follows.

 \begin{theorem}\label{sprqt}
Any symplectic spread of $\PG(3,q)$ gives rise to a maximal partial spread of ${\cal Q}^+(7,q)$ of size $q^{2}+1$.
	\end{theorem}
    \begin{proof}
With the notation introduced in Theorem \ref{spr2}, assume that $\cal D$ is a symplectic spread of $\PG(3,q)$, i.e. every line of $\cal D$ is a line of a fixed symplectic polar space ${\cal W}(3,q)$, and that ${\cal S}\subset {\cal M}_2$. Since the members of $\cal S$ are incident with $\Sigma$ and $\Sigma'$, under the triality map we get a set $E$ of $q^2+1$ points of ${\cal Q}^+(7,q)$ such that each of them is conjugate to two points of ${\cal Q}^+(7,q)$, say $P,P'$, with $P=\Sigma^\tau$ and $P'=\Sigma'^\tau$.
This means that $E$ is a partial ovoid of a Klein quadric ${\cal Q}^+(5,q)$. Since $\cal D$ is symplectic it follows that actually $E$ lies on a parabolic quadric ${\cal Q}(4,q)$ of ${\cal Q}^+(5,q)$.
Then, $\cal S$ is maximal on ${\cal Q}^+(7,q)$ if and only if $E$ is maximal on ${\cal Q}^+(7,q)$ (as a partial ovoid). Indeed, let $P$ be a point of ${\cal Q}^+(7,q)\setminus {\cal Q}(4,q)$. Then $P^\perp$ intersects ${\cal Q}(4,q)$ in either a cone or a hyperbolic quadric or an elliptic quadric. In the former two cases the quadric section contains lines that certainly meet $E$. In the latter case, from \cite{BGS} an elliptic quadric of ${\cal Q}(4,q)$ must meet $E$, and we are done.
\end{proof}

	\begin{remark}
{\rm Assuming that $\cal D$ is Desarguesian, it follows that the triality image of $\cal S$ is actually an elliptic quadric section ${\cal Q}^-(3,q)$ of ${\cal Q}^+(7,q)$. From \cite[15.1 8)]{MA} $\cal S$ corresponds to the $\GF(q)$--linear representation of a regulus of a hyperbolic quadric ${\cal Q}^+(3,q^2)$ and hence we are in the case of Theorem \ref{regulus}.}
	\end{remark}
	
	\begin{remark}
{\rm  It is not difficult to prove that the smallest maximal partial spread of ${\cal Q}^+(7,2)$ has size $5$ and corresponds under triality to an elliptic solid section of ${\cal Q}^+(7,2)$. In the spin representation of $\PSp(6,2)$ such an elliptic solid section corresponds to a maximal partial spread of ${\cal W}(5,2)$. Also, the smallest maximal partial spread of ${\cal W}(5,2)$ has size $5$.} 	
	\end{remark}

Taking into account Lemma \ref{hyperbolic}, one can ask for the maximum number $m$ of symplectic polar spaces $\cW(3,q)$ in a solid such that the projectivity obtained as the product of any two associated symplectic polarities is fixed point free. This is equivalent to ask for the maximum number of $\cW(3,q)$ pairwise intersecting in an elliptic congruence (i.e. a Desarguesian line--spread of a solid), which in turn, using the Klein correspondence, is equivalent to ask for the maximum size of an {\em exterior set} of $\cQ^+(5,q)$, i.e., a set $X$ of points such that each line joining two distinct elements of $X$ has no point in common with $\cQ^+(5,q)$. Since a spread of $\cQ^+(7,q)$ has size $q^3+1$, we have that $m \le q^2+q+1$. On the other hand it is known that $|X| \le q^2+q+1$ and equality holds if and only if $q=2$, see \cite{T1}. Under triality the $q+1$ generators of a hyperbolic Segre variety are mapped to $q+1$ points forming a conic section of ${\cal Q}^+(7,q)$. Therefore there exists no ovoid of $\cQ^+(7,q)$, $q > 2$, that is union of conics sharing two points. However, this fact was already noted in \cite{BCL}.
 
The next result is group--theoretic and gives some information on the stabilizer of a hyperbolic Segre variety ${\cal S}_{1,3}$ of ${\cal Q}^+(7,q)$, $q$ even.

	\begin{proposition}
The stabilizer of a hyperbolic Segre variety ${\cal S}_{1,3}$ of ${\cal Q}^+(7,q)$, $q$ even, is a subgroup of $\Omega(7,q)$ in its spin representation.
	\end{proposition}
	\begin{proof}
Assume without loss of generality that the hyperbolic Segre variety of ${\cal Q}^+(7,q)$ consists of $q+1$ mutually disjoint solids of ${\cal M}_2$. From Remark \ref{grupposh} such solids are permuted by $K\simeq \PSp(2,q)\times \PSp(4,q)$. Under triality they are mapped to $q+1$ points forming a conic section $C$ of ${\cal Q}^+(7,q)$. Since $q$ is even $C$ has a nucleus $N\not\in {\cal Q}^+(7,q)$ that is fixed by $K^\tau$. From \cite{MA} the triality image of the stabilizer of a point not on ${\cal Q}^+(7,q)$ is the spin representation of $\Omega(7,q)$, and we are done.   	
	\end{proof}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%\subsection*{Acknowledgements}
%Thanks to Professor Querty for suggesting the proof of
%Lemma~\ref{lem:Technical}.

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%




\end{document}